Author's response to reviews

Title: Effect of sequential treatment with syndrome differentiation on acute exacerbation of chronic obstructive pulmonary disease and "AECOPD Risk-Window" : a randomized placebo-controlled study

Authors:

Wang Haifeng (wangh_f@126.com)
Li Jiansheng (li_is8@163.com)
Li Suyun (lisuyun2000@yahoo.com.cn)
Yu Xueqing (yxgshi@126.com)
Zhang Hailong (zhanghailong6@126.com)
Wang Zhiwan (zhiwan_w@163.com)
Wu Qiyi (wuqiyi1997@126.com)
Zhang Pankui (zhangpankui@126.com)
Wang Zhongchao (geianna@126.com)
Li Fenglei (leifeng0217@126.com)
Yan Haihong (yhh5858@126.com)

Version: 4 Date: 10 March 2012

Author's response to reviews: see over
Dear Editors-in-Chief

My manuscript entitled as “Effect of sequential treatment with syndrome differentiation on acute exacerbation of chronic obstructive pulmonary disease and"AECOPD Risk-Window": a randomized placebo-controlled study” was revised according to the reviewer’s comments.

As suggest by the reviewer,” After assessing your revised manuscript, the Associate Editor recommended that it be copy-edited to improve the reading of the article. He has very kindly provided a copy of your manuscript with edits he suggests you make to improve the English ”.

Many thanks for your suggestion. We would be grateful and we would address his suggestions in a revised manuscript. All changes have been highlight when the manuscript revised.

I am sorry to bring you so much trouble because of my poor English . Correspondence and phone calls about this paper should be directed to prof. Li Jiansheng at the following address, phone and fax number, and e-mail address:

Institute: Henan University of Traditional Chinese Medicine,

Adress: No.1 Jinshui Road Zhengzhou Henan PR China

Tel: 86-371-65676568

Fax: 86-371-65680028

E-mail: li_js8@163.com

At last, thank you again for your arduous work and instructive advice.